Author's response to reviews

Title: When care situations evoke difficult emotions in nursing staff members: an ethnographic study in two Norwegian nursing homes

Authors:

Anne Marie A.M.S. Sandvoll (annemsa@hisf.no)
Solveig S.H. Hauge (solveig.hauge@hit.no)
Kjell K.K Kristoffersen (kjell.kristoffersen@ui.no)
Ellen Karine E.K.G. Grov (Ellen.karine.grov@hisf.no)

Version: 8  Date: 21 July 2015

Author's response to reviews: see over
Dear Miss Marielette Costoy

BMC Nursing

Førde, July 22th, 2015

Thank you for the editorial comments to our revised manuscript “When care situations evoke difficult emotions in nursing staff members: an ethnographic study in two Norwegian nursing homes”.

We have considered the comments and we are pleased to resubmit a revised article with the following changes (marked in red), made point-by-point according to the editor’s suggestions.

Editorial requirements: Editorial Comments:

1) Authors Contribution: All authors must be mentioned by initials in this section. All authors are now mentioned by initials
2) Please include all authors emails on the title page. We have added all authors’ emails on the title page